# UNIVERSITYOF **BIRMINGHAM** University of Birmingham Research at Birmingham

# **Humanitarianisms in context**

O'sullivan, Kevin; Hilton, Matthew; Fiori, Juliano

DOI:

10.1080/13507486.2015.1117422

Creative Commons: Attribution (CC BY)

Document Version Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

O'sullivan, K, Hilton, M & Fiori, J 2016, 'Humanitarianisms in context', European Review of History/Revue europeene d'Histoire, vol. 23, no. 1-2, pp. 1-15. https://doi.org/10.1080/13507486.2015.1117422

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes

- •Users may freely distribute the URL that is used to identify this publication.
- •Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.
  •User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?)
- •Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

Download date: 25. Apr. 2024



# European Review of History: Revue européenne d'histoire



Date: 24 May 2016, At: 04:39

ISSN: 1350-7486 (Print) 1469-8293 (Online) Journal homepage: http://www.tandfonline.com/loi/cerh20

# **Humanitarianisms in context**

Kevin O'Sullivan, Matthew Hilton & Juliano Fiori

**To cite this article:** Kevin O'Sullivan, Matthew Hilton & Juliano Fiori (2016) Humanitarianisms in context, European Review of History: Revue européenne d'histoire, 23:1-2, 1-15, DOI: 10.1080/13507486.2015.1117422

To link to this article: <a href="http://dx.doi.org/10.1080/13507486.2015.1117422">http://dx.doi.org/10.1080/13507486.2015.1117422</a>

| © 2016 The Author(s). Published by Informa<br>UK Limited, trading as Taylor & Francis<br>Group |
|--|
| Published online: 16 Mar 2016.   |
| Submit your article to this journal $oldsymbol{oldsymbol{\mathcal{G}}}$                        |
| Article views: 422   |
| View related articles 🗗  |
| View Crossmark data ☑  |
|  |

Full Terms & Conditions of access and use can be found at http://www.tandfonline.com/action/journalInformation?journalCode=cerh20



# **Humanitarianisms in context**

Kevin O'Sullivana, Matthew Hiltonb and Juliano Fioric

<sup>a</sup>School of Humanities, National University of Ireland Galway, Galway, Ireland; <sup>b</sup>School of History and Cultures, University of Birmingham, Birmingham, United Kingdom; <sup>c</sup>Save the Children UK, London, United Kingdom

#### **ABSTRACT**

This introduction describes the rapidly expanding history of nonstate humanitarianism in terms of three themes. First, it argues that we should think about humanitarianism less in terms of ruptures or breaks, and focus more on the moments of acceleration and the continuities that shaped that narrative; how the relationships among local, national and international discourses were played out in the shift between imperial and post-colonial worlds, in the dialogue between religious and secular traditions, and in the transformative processes of decolonization, de-regulation and globalization. Second, we suggest the need to re-think the geography of non-state humanitarianism. Drawing attention to the transnational contexts and traditions in which ideas of humanitarianism have been articulated not only adds to our understanding of transnational action and the strength of global civil society beyond the West, we argue, it allows us to better appreciate the myriad languages and practices of humanitarianism employed in a global context. Finally, this introduction also re-visits the guestion of motivation. By looking beyond the state, we argue, we can better understand the variety of motives that shaped the act of giving: from compassion to capturing markets, the search for efficiency, and the construction of local, national and international identities.

#### **ARTICLE HISTORY**

Received 17 July 2015 Accepted 4 November 2015

#### **KEYWORDS**

Humanitarianism; nongovernmental organizations; empire; globalization; Cold War; development; foreign aid

#### Introduction

Humanitarianism is one of the defining features of our contemporary world. It's on our televisions. It's in our newspapers. It's even on our streets: from the tin-rattling collectors in branded t-shirts to the 'chuggers' who ask, 'Can you spare a moment for ...?'. At its forefront are the non-governmental organizations (NGOs) that implore us to give in times of crisis and to remember them in times without. Anyone can have an impact, they promise: 'It's amazing what can be achieved with just a small amount of money from people like you ... Your help will make a real difference to people's lives around the world.' Yet this 'concern for distant strangers' is nothing new. On the missions, in empire, through the international Red Cross movement, and particularly in the aftermath of the two world wars, the last

150 years have witnessed the rise of non-state humanitarianism as a global norm. NGOs have become model transnationalists, conduits for internationalism, and - through their language and actions - interlocutors between their supporters and the outside world. They have also proliferated as humanitarianism has been adapted and articulated in its myriad cultural forms: from Haku-Ai-Sha in 1870s Japan to Islamic Relief's expanding world role.<sup>3</sup>

The articles in this volume describe the rise of humanitarianism since the mid-nineteenth century as a global phenomenon, played out in any number of sites across the world – East and West, North and South. They build on a decade-long boom in historical research on humanitarianism, aid and development that has opened myriad new avenues for understanding the experience of, and motivations for, such altruistic actions. By focusing our attention away from official agencies and towards non-state actors, however, they offer an important corrective to much work on the mechanics of global humanitarianism to date. They prompt us to think again about the relationships between donors and recipients. They ask us to revisit the interactions among local, national and international languages of charitable action. And they lead us in turn to seek out the spaces (physical and metaphorical) in which those narratives took shape.

They do so by addressing three key themes. The first is chronology. Moving away from the all-too-prevalent assertion that non-state humanitarianism is the product of a post-Cold War era, or – at best – of post-Second World War reconstruction in Europe, the articles here point towards a set of more deeply rooted norms that have shaped the contemporary humanitarian world. They draw on previous attempts to describe the period from the mid-nineteenth century in terms of ages or phases of humanitarian action, but ask us to think more deeply about the continuities that have shaped the history of NGOs.<sup>4</sup> Prominent among them are the humanitarian encounters that inform our vision of the outside world. The history of humanitarianism, Rob Skinner and Alan Lester remind us, is intrinsically linked to that of empire - both in its imperial origins and in its attempts to come to terms with the realities of a postcolonial world.<sup>5</sup> Empire was just as important in shaping Indian humanitarian responses to the crisis in Spain in the 1930s as it was to the language and interpretative frameworks employed by West German aid workers operating through Cap Anamur more than four decades later. But empire is not the only constant that shaped the humanitarian past. Once we begin to read the history of non-state humanitarianism in terms of continuities, this volume suggests, we can also begin to think more broadly in terms of the things that humanitarianism could not leave behind: from local to international forms of humanitarian action (as in the relationship between Haku-Ai-Sha and the Japanese Red Cross, for example), or from pre- to post-(First, Second or Cold) war worlds.

Disrupting the traditional narrative in this way also leads us to revisit the geography of non-state humanitarianism. The articles in this volume broaden our understanding of the locations of humanitarian action, by drawing on new histories of NGOs in Asia, Central and Eastern Europe, and the Middle East. The narrative of humanitarianism, they suggest, is not simply one of aid provided by 'the West' to 'the rest'. From its very beginnings, humanitarianism was not only a North-South but also a South-South phenomenon, and - as in the case of Indian aid to Spain - a South-North one as well. Unpacking how and where those interactions were shaped not only adds to our understanding of transnational action and the strength of global civil society beyond the West, it also allows us to better appreciate the myriad languages and practices of humanitarianism employed in a non-Western

context. What motives and moral frameworks inspired them? Did humanitarianism take on a different shape when viewed through a non-Western lens?

The answers to those questions lead us to the third theme of this volume: motivation. Two decades of research into the history of official aid have provided us with a nuanced picture of the dynamics that drive development co-operation, from realist influence-buying to the pursuit of social-democratic principles on the world stage.<sup>6</sup> In recent years we have also developed a deeper understanding of the individual act of giving: as a way 'to continue believing ... in this concept of humanity which presupposes that all human beings are of equal value because they belong to one moral community.<sup>7</sup> By looking beyond the state, however, the essays in this volume show just how varied – and complex – the motives that shape the act of giving can be. The NGOs they describe were expressions of something more than simple compassion. They were conditioned by political interests and a desire to capture markets. They were driven by the search for efficiency. But they were also closely linked to the construction of local, national and international identities. British youth engagement with NGOs, for example, became a way of finding a new role for the nation and its people in the aftermath of empire, and an attempt by a new generation to distance itself from the legacies of empire. Like many of the actors in this volume their donations – in time and money – were the result of what we might term humanitarian patriotism. But they also exhibited a strong sense of what Tamara Myers described in the Canadian case as a 'global helpmate identity': a way of projecting their international good citizenship on to the world stage.8

Together, the essays in this volume make a rich and lively contribution to the growing academic debate on humanitarianism. But they also speak to one additional, overarching theme: the growing interest in history across the contemporary humanitarian sector. In 2011, for example, the Overseas Development Institute (ODI) began work on its 'Global History of Modern Humanitarian Action' project, 'to promote the use of history in the practice and policy-making of humanitarian action. The same year, Médecins sans Frontières (MSF) published Humanitarian Negotiations Revealed, an open - and frequently critical - discussion of its work in the field. 11 That organization has also since made available a rich array of case studies and primary-source materials through its Speaking Out website. 12 Save the Children has also been working with historians to develop a more complete and nuanced understanding of its history and politics that can inform future strategy.<sup>13</sup> And those organizations are not alone in their desire to apply the lessons of the past to the policy-making sphere. Representatives from both ODI and MSF, along with officials from a number of other European NGOs, were integrally involved in the 'Non-state Humanitarianism' research network from which these essays emerged.<sup>14</sup> In April 2014 the network convened a workshop on 'the global and the local in humanitarian action' at Save the Children's London offices, involving participants from Oxfam, Plan International, Christian Aid, ODI, MSF, the European Commission's Humanitarian Aid and Civil Protection department (ECHO), the UN Office for the Co-ordination of Humanitarian Affairs (UNOCHA), Amnesty International, and others. The spirit of those conversations, and the connections they generated, is visible in the essays presented here.

On account of the urgency of emergency response, the humanitarian sector seems to have little time to reflect on the past. Yet campaigning and fundraising narratives often invoke historical moments. Moreover, humanitarian discourse is, to a remarkable extent, built upon golden-era narratives, hagiographies and mythmaking, Whiggish reconstructions, and marketable prophecies about 'unprecedented' and apocalyptic developments in

human affairs. Historically grounded and thematically expansive, the essays in this collection speak to a desire to learn from the past and enrich our understanding in terms of context and process. 15 They explore histories that challenge essentialist and simplistic assumptions about the origins and nature of humanitarian concern. In that sense they should be seen as part of a broader attempt to present history as a method of critical reflection; indeed, one that can be valuable to the humanitarian sector. 16 And in this way, they can serve as an antidote to the treatment of history as a 'useful' commodity or repository of vendible stats and anecdotes.

# Mapping the history of non-state action

Humanitarianism is a broad church. To many in the contemporary aid sector, the term has come to be associated with crisis, intervention and disaster relief. Consequently, NGO structures, fund-raising models and campaign narratives have followed suit, broadly subscribing to the notion of 'humanitarian organizations ... careening from one major emergency to another.' Yet to historians the story is much more complex. When its narrative is read over a 150-year period, humanitarianism becomes 'a complex entity, with many arms and faces'. 18 Its protagonists (including NGOs) saw their roles not in terms of the compartmentalized division of labour but as a range of benevolent actions that defined their engagement with the outside world. The Save the Children Fund, to take just one prominent example, made its name in responding to humanitarian catastrophe in the aftermath of the First World War, but was also one of the leading advocates of the rights of the child and ran a large number of children's homes in Central and Eastern Europe in the inter-war period - each of which it viewed as inseparable parts of its overall mission.<sup>19</sup> It was not alone. As NGOs grew into larger and more complex entities, particularly in the second half of the twentieth century, their activities on the ground were articulated in terms of a broad spectrum of 'humanitarian' activities. In El Salvador, for example, Irish NGO Trócaire's attempts to provide immediate relief to the thousands made refugees by that country's civil war in the early 1980s were closely intertwined with a broader campaign to 'help those suffering the effects of human rights violations'. 20 By the middle of the following decade, the 'right to development' had become integral to mainstream NGO speak.

It follows that we should read the historiography of NGOs in terms of a broader shift in our understanding of the humanitarian impulse over the past decade. Histories of official aid have led the way by mapping the patterns of motivations and international relations that drove the aid industry, as well as aid's implications for global governance.<sup>21</sup> The recent boom in human-rights history has added considerably to that debate, drawing out the discussion of where and how our humanitarian sensibilities are formed, and tracing the pursuit of those ideals in a broadly transnational context.<sup>22</sup> It has also underlined the holistic nature of humanitarian discourse and action. As scholars continue to chip away at the edifice of the rights narrative, it becomes ever clearer that its histories are inseparable from our understanding of what it means to feel compassion and how that solidarity has been expressed: what Frederick Cooper termed 'the relationship of individual, state, and humanity'. The historically grounded, but inter-disciplinary focused Humanity: An International Journal of Human Rights, Humanitarianism and Development has best embodied that shift. Since its establishment in 2010 that journal has emphasized the entangled nature of rights, aid, development and disaster relief. The baton of holisticism has in turn been taken up elsewhere,

including in one recent article that argues for a hybrid 'humanitarian rights' category when describing the International Committee of the Red Cross's (ICRC) intervention in Soviet Russia in the aftermath of the First World War.<sup>24</sup>

Historians of non-state humanitarianism have benefited enormously from that evolving discourse. Michael Barnett's Empire of Humanity: A History of Humanitarianism (2011) remains the most obvious marker of an explosion of interest in the subject that has occurred since the beginning of the 2010s, but his work has been accompanied by a rapidly expanding body of articles, monographs, edited collections and a vibrant postgraduate research community that have taken us a considerable distance in our quest to unravel the dynamics of humanitarian action.<sup>25</sup> The availability of new archives and online resources has greatly assisted in that process. New narratives and research foci followed, from studies of aid's recipients to the technologies and spaces of humanitarian aid.<sup>26</sup> With them come a number of key questions that inform this volume, all of which focus on the languages of humanitarianism and how they have been promoted, adapted and contested in their various contexts. What are the different levels of historical engagement between local actors and transnational networks of humanitarianism? Have there been specific local, national and regional contributions to these global debates? Where does the stimulus come for change? Can we understand historical shifts in humanitarian discourse in terms of key 'moments' of humanitarianism? What are they?

In answering those questions, the focus on non-state actors here is significant. The essays in this volume are framed within a methodological shift that began with Akira Iriye's call in 1999 to re-think the twentieth century as 'a century of NGOs'. The commitment that those organizations exhibited 'to activism derived from a moral conception of the world, their humanitarianism, and their support of human (and animal) rights', Iriye argued, made them an important barometer of the 'core values' of that period. And though he wrote primarily of the United States, Iriye's emphasis on non-state actors proved an inspiration for the scholars that followed. Global historians looked to NGOs as key building blocks of an interdependent world: 'Stumbling, fumbling, fudging, and nudging, NGOs as a social movement are a major, almost dominant factor in present-day globalization. As someone said of religion, if it didn't exist, we might have to invent it.'28 Others traced the origins and structure of transnational civil society through the story of non-state action.'29 NGOs offer a window onto a world outside the Cold War and great politics and a way of understanding the 'ordinariness' of political action in the twentieth-century world.'30

This volume follows in their path. The better we understand the actions of NGOs and the kind of 'people-to-people' action that they precipitated, it suggests, the better we can appreciate the myriad layers that make up our response to poverty, disaster and the pursuit of justice in the outside world. But its impact is not limited to the humanitarian sphere. The articles in this volume allow us to re-visit a number of established fields of history by looking anew at the patterns of interaction and exchange that shaped them. They use the lens of humanitarianism to revisit debates about empire, from the ways in which the discourses of 'imperial humanitarianism' and 'neo-humanitarianism' were shaped, to the role of humanitarianism in shaping identity in a post-colonial world.<sup>31</sup> And they ask us to think again about the nature of globalization and global governance: how and where power is wielded, and how and where the relationships that shaped those processes were formed.

Woven into their narratives are stories of internationalism, modernization, development and the evolution of global civil society that move beyond a narrative of a Western-imposed

rationality and towards a more flexible understanding of the kinds of transnational exchanges that have shaped the modern world. By bringing these various case studies into dialogue with each other, they draw attention to the locations and traditions in which ideas of non-state humanitarianism have been articulated, and how these interact with their local, national and international contexts. In doing so the essays here also allow us to develop the 'century of NGOs' concept and to move towards a more nuanced understanding of the good and bad of non-state action - all the while avoiding a zero-sum game approach that assumes the rise of NGOs necessitated an equally diminished role for the state.

# Chronology

The 1940s are widely regarded as a significant moment of change in the history of humanitarianism.<sup>32</sup> The establishment of the United Nations Relief and Rehabilitation Administration (UNRRA) ushered in a new era of official intervention that dwarfed the previous efforts of the voluntary sector. United States President Harry Truman's famous Point Four declaration - part of his inauguration speech in 1949 - introduced the concept of 'under-development' and marked the beginning of the era of official development aid. The decade was an equally significant one for NGOs; humanitarian crisis in Europe created new opportunities for nonstate actors, while the expansion of the UN agencies into the Middle East and Asia later in the decade prompted NGOs to follow suit. For existing organizations those developments precipitated a massive expansion in their activities – both in terms of the attention they received, and of their ambition and scope. Most significantly, however, the war and its aftermath also provided the impetus for new actors to emerge, among them Oxfam and Christian Aid in Britain, and CARE and Catholic Relief Services in the United States,

Yet this emphasis on the 1940s as a period of change has tended to sideline important continuities that have shaped humanitarian thinking and practice since at least the mid-nineteenth century. The essays in this volume demonstrate significant institutional continuities in non-state organization from that period onwards. Not only were imperial connections and mindsets allowed to continue, but the interwar period also proved to be something of an experimental moment for the sort of transnational solidarities that later became far more common. Indeed, in terms of the history of unofficial humanitarianism, the moment after the First World War appears just as important as that after the Second, as demonstrated by the essay by Friederike Kind-Kovacs in this volume. It was then that organizations such as Save the Children appeared, seeking to cross official policies on intervention in order to relieve the suffering of those civilians caught up in conflict. It was then that communities of officials and technicians were formed that would later lead the humanitarian sector. And it was equally visible then that humanitarians did not operate independently of the global power struggles that helped to shape and direct their interventions. What Branden Little termed 'the humanitarian war' lasted until well into the 1920s and 'endures in both the prevailing concepts of humanitarian assistance and disaster relief and the vast international aid regime today - the indisputable by-products of those catalytic endeavours.'33

Rather than replace one genesis moment (the 1940s) with another (the First World War and its aftermath), however, the essays in this volume suggest viewing humanitarianism in terms of moments of acceleration. We should read its history not as a series of sudden right turns, but instead see it in terms of bursts of activity that refreshed the sector while carrying with them the baggage of what had come before. Reading the history of non-state

humanitarianism in this way, the intersection of imperial humanitarianism and Henri Dunant's brand of Red Cross internationalism becomes the first such phase, the post-First World War period another, and the 1940s the third moment of significant expansion for the non-state sector. Since then, humanitarianism has undergone two further moments of acceleration. The fourth phase came in the 1968–85 period – the 'NGO moment' between the Biafran humanitarian crisis and the global fund-raising extravaganza of Live Aid – when the rise of global media, the creation of a global development industry from above, and the immediacy of crisis in the decolonized world fostered a proliferation of NGOs, allowing them to become the primary manifestations of global concern for the less-well-off.<sup>34</sup> The post-Cold War period was the fifth, when the immediacy of the bipolar conflict had dissipated and prompted the search for new forms of intervention in a rapidly changing world.

Taking that approach a step further, thinking in terms of accelerations underlines the continuities that have shaped non-state humanitarianism since the mid-nineteenth century. Each of our five moments of acceleration was defined not only by the changes it brought to the international NGO community, but also by the people, languages and practices of humanitarianism that endured. The continuities from empire are many. Humanitarianism follows paths established by centuries of imperial activity, clings to the vernacular of empire in shaping how we 'imagine' the outside world, and adapts the organizing frameworks of 'civilization' to our efforts to understand inequality and poverty in a globalized era. In keeping with those imperial sensibilities, contemporary NGOs have adopted what Nandita Dogra terms 'a double logic of "difference" and "oneness" ... to show the global poor as different and distant from the [developed world] and yet like us by virtue of their humanity.'35

And such continuity is also visible in other parts of the sector. The 'professionalization' of non-state humanitarianism, for example, is often described as a product of the rapid expansion of the sector in the 1970s and 1980s, the rise of 'development studies', and the transformation of 'charity' into an 'aid industry'. Yet if we look more closely at the culture and managerial structures of non-state actors over the nineteenth and twentieth centuries, a number of patterns emerge. The essays in this volume – not least Heike Wieters's essay on CARE and Daniel Maul's work on the Quakers – remind us of the capitalist roots that Thomas Haskell ascribed to humanitarianism. They lead us to think again about where and when the sector 'professionalized', while reminding us, for example, of the Christian missionaries who had worked and trained for decades in the areas of education and health. And they also help us to understand better how deeply imbued humanitarianism has been with the spirit of entrepreneurship: how non-state actors 'managed' their 'consumers', how they engaged in public-private partnerships, and how and where they embraced the tropes of managerialism.

# Geography

In describing humanitarianism in those terms, however, we must be careful not to fall into the trap of reading the history of non-state humanitarianism solely in terms of the imposition of a Western world order. Recent work on transnationalism and global civil society has destabilized our traditional assumptions about the flows that constitute the international system.<sup>37</sup> The essays in this volume extend that process, shifting the focus away from the West to sites of interconnection that appeared across less expected routes. The Red Cross, for example, was already a global phenomenon by the end of the nineteenth century, inflected

with the languages and particularities of its local societies. South-South and South-North transfers were also much more prominent than the recent 'discovery' of 'new' Southern donors suggests. Those narratives open new ground for understanding the practical and intellectual structures of humanitarianism. NGOs, as the essays here by Maria Framke (on Indian assistance to Spain) and Frank Käser (on the Red Cross in Japan) tell us, were products of something more than a Western imagining of compassion. They were shaped and re-shaped according to the local contexts and traditions in which humanitarianism operated.

The same narratives also warn us of the dangers of privileging certain patterns of action and exchange. While recent studies of the Cold War have underlined the significance of the bipolar conflict in shaping global power dynamics, the turn to non-state actors has demonstrated the rich transnational flows of ideas and institutions, care and compassion that operated alongside, and often on the edges of, its many borders and barriers. Akira Iriye accorded this 'global community' of inter-governmental organizations and NGOs great influence in shaping a world of activity 'outside the drama of the Cold War'. Humanitarian organizations were at its forefront. NGOs worked alongside the UN agencies in shaping the fields of development, human rights and disaster relief. They drew individuals - North and South - into new conversations and collaborations that were not easily described within the boundaries of the bipolar conflict. And, in turn, they also gave people a sense that they belonged: to a broader conception of 'humanity', but also to 'an imagined community of solidarity activists' committed to building a better world.<sup>39</sup>

Yet the kind of 'people-to-people' action that humanitarian NGOs pursued was not always the benign process that their supporters would have us suppose. Understanding how and where these organizations operated helps us to unpack what Frederick Cooper termed 'the lumpiness of cross-border connections' and the power that they conceal. 40 The continuities that shaped humanitarian action are significant in that sense: from the enduring language and practices of empire, to the actors (such as colonial officials) that transferred their involvement in one field into the arena of humanitarian aid, its stories underline the crucial role of humanitarian NGOs in global governance.<sup>41</sup> Equally important are the (often blurry) dividing lines between official and unofficial agencies in the humanitarian sector, as Wieters's essay on CARE highlights. NGOs are constantly engaged in debates as to whether certain organizations have become 'too close for comfort' to their donors and have thus lost their independence of thought and action.<sup>42</sup> They have also raised questions about what constitutes a 'state' and 'non-state' actor. The stories told here enable us to glimpse the ever-changing dynamics between non-state and official humanitarianism, especially as NGOs have continued to grow and expand their scopes and fields of operation. As Sending and Neumann put it, those organizations were 'not brought into the fold and as such deprived of their autonomy and as recognized representatives of "civil society". Rather, it is precisely their status as actors that possess expertise central to the task of governing ... that make them key subjects of, and allies in, governmental tasks.'43

That assertion also has important implications for how we understand the impact of NGOs on the nation-state. It reminds us first of all that increased non-state activity did not always lead to a corresponding diminution of power at the level of the nation-state, nor did the re-assertion of state power necessarily lead non-state actors to fall away. Once we begin to pursue that line of argument, it becomes possible to see NGOs not simply as alternatives to the official apparatus, but as carriers of the values of, and providers of services for, the state. That process was visible in empire – as Skinner and Lester argue: 'The long-distance webs of concern spun by humanitarians within empire have always been intrinsic to the politics both of empire itself and of nation-state foundation.'<sup>44</sup> Non-state humanitarians not only acted according to altruistic motives; those same actions reinforced and helped to constitute the 'civilizing' mission of the imperial project. Their influence endured in the aftermath of empire too. NGOs elaborated the response to decolonization by providing an outlet for compassion, the expression of interest – and enduring (if by then largely informal) forms of influence. By the 1980s they had also become agents for new forms of pseudo-imperial control, not least the advance of neoliberalism and the deregulating policies of the World Bank and the International Monetary Fund in Africa, Asia and further afield.<sup>45</sup> The history of the interconnections between the official and unofficial spheres is a messy one, as Peter Gatrell's essay shows. If ever a subject demanded the need for further research, it lies in that fuzzy and ever-shifting boundary – financial, ideological, institutional, personal – between the governmental and non-governmental worlds.

#### **Motivations**

The emphasis placed here on seeking out the nature of power also leads us to investigate the complex reasons why we give aid - and why we direct it through NGOs. There is a long-standing critique that charity says more about the donor than the recipient, that it alleviates the guilt of the advantaged more than the poverty of the poor; what Didier Fassin termed 'the response made by our societies to what is intolerable about the state of the contemporary world'.46 Too often, however, these judgements are made on assumptions, and we actually know relatively little about the motivations of donors (whether or time or money). Many of the articles here, especially by Anna Bocking-Welch on Youth Against Hunger and Patrick Merziger on Cap Anamur, by either investigating the nature of giving or by looking at who is giving (and to what), help to uncover the range of reasons why people support and get involved with humanitarian agencies, as well as what their support says about their view of the world. They start from the premise that NGOs have always had to make very deliberate decisions as to how their work fits in with their goals and beliefs. More than official agencies, non-state actors are prone to soul-searching struggles over the meanings of their interventions. In this sense, they become a particularly rich site for the examination of the ever-changing sphere and meaning of humanitarianism. To act and to intervene involves discussions over their proper roles and the locations of aid. The very fact that no new humanitarian action is ever the same as that which came before imposes a constant dialogue upon those organizations: whether they can or should act.

Those discussions, in turn, are shaped by the ever-shifting spaces of humanitarian thought and practice. To understand the relationships among local, national and international discourses of humanitarianism, we need to explore how those interactions were played out in the transition between imperial and post-colonial worlds, in the dialogue between religious and secular traditions, and in the bonds between non-state actors and their donor publics. The essays in this volume throw light on two of those spaces. First, they underline the role played by NGOs in reinforcing the kind of 'internationalist project' that Emily Baughan ascribed to Save the Children in the 1920s.<sup>47</sup> In that sense, humanitarianism became a way of shaping the kind of world that the donors wished to mould. NGOs pursued their individual visions of 'welfare', 'development', 'progress' and 'modernization' in the name of 'trying to remake humanity to fit global norms'. Religious organizations, for example,

mixed humanitarianism with a 'spiritual internationalism' that emphasized reconciliation through personal religious practice. Cap Anamur's 'radical humanism' in the 1980s took that process to another level, by shifting attention away from the territorial 'occupation' involved in development and towards a tightly organized form of control justified by the imperative of humanitarian relief.

Second, the essays here also allow us to think about the intersections between humanitarianism and identity. Non-state actors played an important role in shaping individual and collective senses of belonging by blurring the lines between humanitarianism's local, regional, imperial and international contexts. The 'humanitarian patriotism' involved in providing aid from India during the Spanish Civil War could only be understood within the context of empire and the local and international traditions from which it derived. Egyptian Red Crescent volunteers in the 1950s were also involved in a project that had strong local, regional and international dimensions, while their counterparts in Jami'at al' Islam pursued an anti-Communist and anti-colonial rhetoric from their base in Munich. In post-imperial Britain the re-making of young people into volunteers, global citizens and financial donors was inseparable from domestic concerns about the country's future expressed through youth. In each case the act of giving involved individuals in decisions that go far beyond our traditional understanding of why we donate, reminding us that while humanitarianism allows us 'to hold onto the shreds of [our] humanity', it also touches many other elements of our individual and collective identity.<sup>49</sup>

### **Conclusion**

The themes highlighted in this special issue – chronology, geography, motivation – are too broad for definitive answers to be found in what follows. The history of humanitarianism remains a fledgling if vibrant subject. The articles below constitute empirically rich case studies that will in turn go on to become expanded monographs and eventually syntheses of rapidly expanding bibliographies. They make a contribution to what is sure to become a flood of new work emerging from a new generation of scholars concerned, at the widest level, with major questions about the interconnected nature of the modern world. Endeavours to understand humanitarianism can contribute to the demystification of those interconnections, complementing and usefully complicating the significant body of work already undertaken on the darker aspects of modernity's global exchanges. What will emerge is further evidence for what scholars just a decade ago recognized as the rise of the NGOs. But rather than tracing the uncomplicated rise of a force for good in the world, new work on humanitarianism – in line with the articles presented here – will problematize the nature and the effects of such organizations. In doing so, as with work on human rights, it will present the humanitarians not so much as angels of mercy, but as powerful forces that raise fundamental questions about the development and dynamics of modern humanity.

#### **Notes**

- Oxfam International, "Donate." 1.
- 2. Skinner and Lester, "Humanitarianism and Empire," 732.
- The Overseas Development Institute's "Global History of Modern Humanitarian Action" project (2011-15) has been at the forefront in highlighting the long history of non-Western



- forms of humanitarianism; http://www.odi.org/projects/2547-global-history-modern-humanitarian-action-moving-forward-hpg (Accessed 23 October 2014).
- 4. See, for example, Michael Barnett's "ages" of humanitarianism: Barnett, *Empire of Humanity*; and Johannes Paulmann's "conjunctures" approach: Paulmann, "Conjunctures in the History of International Humanitarian Aid."
- 5. Skinner and Lester, "Humanitarianism and Empire."
- See, for example, Lancaster, Foreign Aid; Lumsdaine, Moral Vision; Noël and Thérien, "From Domestic to International Justice;" Olesen, Pharo, and Paaskesen, Saints and Sinners; and Stokke, Western Middle Powers.
- 7. Fassin, Humanitarian Reason, 252.
- 8. Myers, "Local Action and Global Imagining." On Britain, see Bocking-Welch, "Imperial Legacies and Internationalist Discourses."
- 9. See, for example, Davey and Scriven, "Aid in the Archives."
- 10. Overseas Development Institute, "Global History of Modern Humanitarian Action."
- 11. Magone, Neuman, and Weissman, Humanitarian Negotiations Revealed.
- 12. Médecins sans Frontières, "MSF Speaking Out," url: http://speakingout.msf.org (Accessed 23 October 2014).
- 13. See, for example, Baughan and Fiori, "Save the Children, the Humanitarian Project, and the Politics of Solidarity."
- 14. For details, see the network website, http://nonstatehumanitarianism.com (Accessed 23 October 2014).
- 15. Woolcock, Szreter and Rao ("How and Why History Matters," 12–13) see these as the two key lessons of history for the contemporary humanitarian sector. Similarly, Borton and Davey ("History and Practitioners," 154) suggest that a "greater of use of history" can have "contextual and strategic" benefits for the humanitarian sector. See also Davey, "Humanitarian History in a Complex World."
- 16. Ramos Pinto and Taithe ("Doing History in Public?," 11) suggest that it is as a "mode of thinking about and questioning the world, both past and present" that history can have "impact."
- 17. Barnett and Weiss, "Humanitarianism," 1.
- 18. Mazlish, Idea of Humanity, 72.
- 19. Save the Children's early history is among the best documented of the humanitarian NGOs. See, for example, Baughan, "Every Citizen of Empire;" Boucher, "Cultivating Internationalism; and Mahood and Satzewich, "The Save the Children Fund."
- 20. "Trócaire launches annual Lenten Campaign," One World, April 1982.
- 21. On foreign aid, see fn. 6, above. On the history of development, modernisation and global poverty, see Cullather, *Hungry World*; Ekbladh, *Great American Mission*; Escobar, *Encountering Development*; Frey, Kunkel, and Unger, *International Organisations and Development*; and Rist, *History of Development*.
- 22. See, for example, Hoffmann, *Human Rights*; Iriye, Goedde, and Hitchcock, *Human Rights Revolution*; and Moyn, *Last Utopia*.
- 23. Cooper, "Social Rights and Human Rights," 473.
- 24. Lowe, "Humanitarianism and National Sovereignty," 654.
- 25. Among the most recent publications are Cabanes, *Great War and the Origins of Humanitarianism*; Forclaz, *Humanitarian Imperialism*; Little, "Humanitarianism in the Era of the First World War;" Laqua, "Ideas, Practices and Histories;" Mann, *From Empires to NGOs*; and Watenpaugh, *Bread from Stones*.
- 26. See, for example, Gatrell, *Making of the Modern Refugee*; Lester and Dussart, *Colonisation and the Origins of Humanitarian Governance*; and Scott-Smith, "How Projects Rise and Fall."
- 27. Iriye, "A Century of NGOs," 435.
- 28. Mazlish, New Global History, 50.
- 29. See, for example, Davies, NGOs; and Alston and Laqua, "Transnational Solidarities."
- 30. Hilton, "Politics is Ordinary;" and Hilton, McKay, Crowson, and Mouhot, Politics of Expertise.

- 31. The concepts of "imperial humanitarianism" and "neo-humanitarianism" are developed in Barnett, Empire of Humanity.
- 32. The 1940s has received considerable scholarly attention in recent years. See, for example, Cohen, In War's Wake; Reinisch, "Relief in the Aftermath of War;" Reinsich, "Internationalism in Relief;" and Reinisch, "Auntie UNRRA."
- 33. Little, "An Explosion of New Endeavours," 13.
- 34. For an introduction to the "NGO moment" concept, see O'Sullivan, "Humanitarian Encounters."
- 35. Dogra, Representations of World Poverty, 3.
- 36. Haskell, "Capitalism, part 1;" and Haskell, "Capitalism, part 2."
- 37. See, for example, Arsan, Lewis, and Richard (eds), "Roots of Global Civil Society;" and Hilton and Mitter, "Transnationalism."
- 38. Iriye, Global Community, 62.
- 39. The concept of "an imagined community of solidarity activists" is developed in Thörn, Anti-Apartheid.
- 40. Cooper, Colonialism in Question, 95.
- 41. See, for example, the kinds of exchanges described in Connelly, Fatal Misconception; Cullather, Hungry World; and Duffield, Development, Security and Unending War.
- 42. Hulme and Edwards, NGOs, States and Donors.
- 43. Sending and Neumann, "Governance to Governmentality," 668.
- 44. Skinner and Lester, "Humanitarianism and Empire," 731.
- 45. For an introduction to the Bank's relationship with NGOs, see Cleary, "World Bank and NGOs."
- 46. Fassin, Humanitarian Reason, 252.
- 47. Baughan, "Every Citizen of Empire," 131.
- 48. Connelly, Fatal Misconception, 380.
- 49. Mazlish, Idea of Humanity, 69.

## **Funding**

The articles in this special issue emerged from the 'Non-State Humanitarianism' international research network, generously funded by an Arts and Humanities Research Council Research Networking Grant [AH/K002805/1], with additional support from the Irish Research Council (New Foundations Grant), the University of Birmingham, National University of Ireland, Galway, Zentrum für Zeithistorische Forschung Potsdam, and Save the Children UK.

### **Notes on contributors**

Kevin O'Sullivan is a lecturer in history at the National University of Ireland, Galway, and an honorary research fellow at the University of Birmingham. His first book, Ireland, Africa and the End of Empire: Small State Identity in the Cold War, 1955-75, was published by Manchester University Press (2012), and he has written most recently on the history of humanitarian NGOs in *Humanity* (2015), the Journal of Genocide Research (2014), and in Frey et al. (eds), International Organizations and Development, 1945-1990 (Palgrave Macmillan, 2014). His current project examines the social and political phenomena that shaped the rapid expansion of the global humanitarian NGO sector between the 1960s and the 1980s.

Matthew Hilton is Professor of Social History at the University of Birmingham. His most recent books include Prosperity for All: Consumer Activism in an Era of Globalisation (Cornell, 2009) and The Politics of Expertise: How NGOs shaped Modern Britain (Oxford, 2013). He has co-edited several collections of essays including The Ages of Voluntarism (OUP, 2011) and Transnationalism and Contemporary Global History (P&P/OUP, 2013). He sits on the editorial boards of Contemporary British History, and the Journal of Consumer Culture and is the editor of Past and Present. He is currently working on the history of British development charities.

*Juliano Fiori* is senior humanitarian affairs advisor at Save the Children UK, where he is head of a team within the humanitarian department dedicated to critical analysis and strategic planning. His current work focuses on solidarity and the politics of aid, humanitarian governance, and the limits of humanitarian effectiveness.

## References

- Alston, Charlotte, and Daniel Laqua, eds. "Transnational Solidarities and the Politics of the Left, 1890–1990." Special issue of *European Review of History* 21, no. 4 (2014).
- Arsan, Andrew, Su Lin Lewis, and Anne-Isabelle Richard, eds. "The Roots of Global Civil Society and the Interwar Moment." Special issue of *Journal of Global History* 7, no. 2 (2012).
- Barnett, Michael. *Empire of Humanity: A History of Humanitarianism*. Ithaca, NY: Cornell University Press, 2011.
- Barnett, Michael, and Thomas G. Weiss. "Humanitarianism: A Brief History of the Present." In *Humanitarianism in Question: Politics, Power, Ethics*, edited by Michael Barnett and Thomas G Weiss, 1–48. Ithaca, NY: Cornell University Press, 2008.
- Baughan, Emily. "Every Citizen of Empire Implored to Save the Children! Empire, Internationalism and the Save the Children Fund in Inter-War Britain." *Historical Research* 86, no. 231 (2013): 116–137.
- Baughan, Emily, and Juliano Fiori. "Save the Children, the Humanitarian Project, and the Politics of Solidarity: Reviving Dorothy Buxton's Vision". *Disasters* 39, no. 2 (2015): 129–145.
- Bocking-Welch, Anna. "Imperial Legacies and Internationalist Discourses: British Involvement in the United Nations Freedom from Hunger Campaign, 1960–70." *Journal of Imperial and Commonwealth History* 40, no. 5 (2012): 879–896.
- Borton, John, and Eleanor Davey. "History and Practitioners: The Use of History by Humanitarians and the Potential Benefits of History to the Humanitarian Sector." In *The Impact of History? Histories at the Beginning of the Twenty-First Century*, edited by Pedro Ramos Pinto and Bertrand Taithe, 153–168. Abingdon: Routledge, 2015.
- Boucher, Ellen. "Cultivating Internationalism: Save the Children, Public Opinion and the Meaning of Child Relief." In *Brave New World: Imperial and Democratic Nation-Building in Britain Between the Wars*, edited by Laura Beers and Geraint Thomas, 169–188. London: Institute of Historical Research, 2011.
- Cabanes, Bruno. The Great War and the Origins of Humanitarianism, 1918–1924. Cambridge: Cambridge University Press, 2014.
- Cleary, Seamus. "The World Bank and NGOs." In *The Conscience of the World: The Influence of Non-Governmental Organisations in the UN System*, edited by Peter Willetts, 63–97. Washington, D.C.: Brookings Institution, 1996.
- Cohen, Gerard Daniel. In War's Wake: Europe's Displaced Persons in the Postwar Order. Oxford: Oxford University Press, 2012.
- Connelly, Matthew. Fatal Misconception: The Struggle to Control World Population. Cambridge, MA: Harvard University Press, 2008.
- Cooper, Frederick. Colonialism in Question: Theory, Knowledge, History. Berkeley, CA: University of California Press, 2005.
- Cooper, Frederick. "Social Rights and Human Rights in the Time of Decolonisation." *Humanity* 3, no. 3 (2012): 473–492.
- Cullather, Nick. *Hungry World: America's Cold War Battle Against Poverty in Asia*. Cambridge, MA: Harvard University Press, 2010.
- Davey, Eleanor. "Humanitarian History in a Complex World." ODI Humanitarian Policy Group policy brief no. 59 (May 2014), http://www.odi.org/publications/8407-humanitarian-history-past-aid (Accessed 23 Oct. 2014).
- Davey, Eleanor, and Kim Scriven, eds. "Aid in the Archives: Academic Histories for a Practitioner Audience." *Disasters* 39, no. 2 (2015): 113–260.
- Davies, Thomas. NGOs: A New History of Transnational Civil Society. London: Hurst & Co., 2013.

Dogra, Nandita. *Representations of World Poverty: Aid, Development and International NGOs.* London: I. B. Tauris, 2012.

Duffield, Mark. Development, Security and Unending War: Governing the World of Peoples. Cambridge: Polity Press, 2007.

Ekbladh, David. *The Great American Mission: Modernisation and the Construction of an American World Order.* Princeton, NJ: Princeton University Press, 2010.

Escobar, Arturo. *Encountering Development: The Making and Unmaking of the Third World.* Princeton, NJ: Princeton University Press, 1995.

Fassin, Didier. *Humanitarian Reason: A Moral History of the Present*. Berkeley, CA: University of California Press, 2012.

Forclaz, Amalia Ribi. Humanitarian Imperialism: The Politics of Anti-Slavery Activism. Oxford: Oxford University Press, 2015.

Frey, Marc, Sönke Kunkel, and Corinna Unger eds. *International Organizations and Development*, 1945–1990. Basingstoke: Palgrave Macmillan, 2014.

Gatrell, Peter. The Making of the Modern Refugee. Oxford: Oxford University Press, 2013.

Haskell, Thomas L. "Capitalism and the Origins of the Humanitarian Sensibility, Part 1." *American Historical Review* 90, no. 2 (1985): 339–361.

Haskell, Thomas L. "Capitalism and the Origins of the Humanitarian Sensibility, Part 2." *American Historical Review* 90, no. 3 (1985): 547–566.

Hilton, Matthew. "Politics is Ordinary: Non-Governmental Organisations and Political Participation in Contemporary Britain." *Twentieth-Century British History* 22, no. 2 (2011): 230–268.

Hilton, Matthew, and Rana Mitter, eds. "Transnationalism and Contemporary Global History." *Past and Present* suppl. 8 (2013).

Hilton, Matthew, James McKay, Nicholas Crowson, and Jean-François Mouhot. *The Politics of Expertise: How NGOs Shaped Modern Britain*. Oxford: Oxford University Press, 2013.

Hoffmann, Stefan-Ludwig ed. *Human Rights in the Twentieth Century*. Cambridge: Cambridge University Press, 2011.

Hulme, David, and Michael Edwards eds. NGOs, States and Donors: Too Close for Comfort? London: Macmillan, 1997.

Iriye, Akira. "A Century of NGOs." Diplomatic History 23, no. 3 (1999): 421-435.

Iriye, Akira. Global Community: The Role of International Organisations in the Making of the Contemporary World. Berkeley, CA: University of California Press, 2002.

Iriye, Akira, Petra Goedde, and William I. Hitchcock eds. *The Human Rights Revolution: An International History*. Oxford: Oxford University Press, 2012.

Lancaster, Carol. Foreign Aid: Diplomacy, Development, Domestic Politics. London: University of Chicago Press, 2007.

Laqua, Daniel, ed. "Ideas, Practices and Histories of Humanitarianism." Special issue of *Journal of Modern European History* 12, no. 2 (2014).

Lester, Alan, and Fae Dussart. Colonisation and the Origins of Humanitarian Governance: Protecting Aborigines Across the Nineteenth-Century British Empire. Cambridge: Cambridge University Press, 2014

Little, Branden, ed. "Humanitarianism in the Era of the First World War." Special issue of *First World War Studies* 5, no. 1 (2014).

Little, Branden. "An Explosion of New Endeavours: Global Humanitarian Responses to Industrialised Warfare in the First World War." *First World War Studies* 5, no. 1 (2014): 1–16.

Lowe, Kimberly A. "Humanitarianism and National Sovereignty: Red Cross Intervention on Behalf of Political Prisoners in Soviet Russia." *Journal of Contemporary History* 49, no. 4 (2014): 652–674.

Lumsdaine, David H. Moral Vision in International Politics: The Foreign Aid Regime, 1949–1989. Princeton: Princeton University Press, 1993.

Magone, Claire, Michaël Neuman, and Fabrice Weissman eds. *Humanitarian Negotiations Revealed:* The MSF Experience. London: Hurst & Co., 2011.

Mahood, Linda, and Vic Satzewich. "The Save the Children Fund and the Russian Famine of 1921–23: Claims and Counter-Claims About Feeding 'Bolshevik' Children." *Journal of Historical Sociology* 22, no. 1 (2009): 55–83.

Mann, Gregory. From Empires to NGOs in the West African Sahel: The Road to Nongovernmentality. Cambridge: Cambridge University Press, 2015.

Mazlish, Bruce. The New Global History. Abingdon: Routledge, 2006.

Mazlish, Bruce. The Idea of Humanity in a Global Era. New York: Palgrave Macmillan, 2009.

Moyn, Samuel. *The Last Utopia: Human Rights in History*. Cambridge, MA: Harvard University Press, 2010.

Myers, Tamara. "Local Action and Global Imagining: Youth, International Development, and the Walkathon Phenomenon in Sixties' and Seventies' Canada." *Diplomatic History* 38, no. 2 (2014): 282–293.

Noël, Alain, and Jean-Phillippe Thérien. "From Domestic to International Justice: The Welfare State and Foreign Aid." *International Organization* 49, no. 3 (1995): 523–553.

Olesen, Thorsten B., Helge Ø. Pharo, and Kristian Paaskesen, eds. Saints and Sinners: Official Development Assistance and its Dynamics in a Historical and Comparative Perspective. Oslo: Akademika Forlag, 2013.

O'Sullivan, Kevin. "Humanitarian Encounters: Biafra, NGOs and Imaginings of the Third World in Britain and Ireland, 1967–70." *Journal of Genocide Research* 16, no. 2–3 (2014): 299–315.

Overseas Development Institute. "Global History of Modern Humanitarian Action." http://www.odi. org/projects/2547-global-history-modern-humanitarian-action-moving-forward-hpg (Accessed 23 October 2014).

Oxfam International. "Donate." http://www.oxfam.org:8000/en/donate (Accessed 17 October 2014). Paulmann, Johannes. "Conjunctures in the History of International Humanitarian Aid during the Twentieth Century." *Humanity* 4, no. 2 (2013): 215–238.

Pinto, Ramos. "Pedro, and Bertrand Taithe, "Doing Public History? Historians in the Age of Impact"." In *The Impact of History? Histories at the Beginning of the Twenty-First Century*, edited by Pedro Ramos Pinto and Bertrand Taithe, 1–20. Abingdon and New York: Routledge, 2015.

Reinisch, Jessica, ed. "Relief in the Aftermath of War." Special issue of *Journal of Contemporary History* 43, no. 3 (2008): 371–551.

Reinsich, Jessica. "Internationalism in Relief: The Birth (and Death) of UNRRA." *Past and Present suppl.* 6 (2011): 258–289.

Reinisch, Jessica. "Auntie UNRRA' at the Crossroads." Past and Present suppl. 8 (2013): 70-97.

Rist, Gilbert. *The History of Development: From Western Origins to Global Faith*. London: Zed Books, 2011.

Scott-Smith, Tom. "How Projects Rise and Fall: The Lifecycle of a Dietary Modernisation Scheme." *Development in Practice* 24, no. 7 (2014): 785–796.

Sending, Ole Jacob, and Iver B. Neumann. "Governance to Governmentality: Analysing NGOs, States, and Power." *International Studies Quarterly* 50, no. 3 (2006): 651–672.

Skinner, Rob, and Alan Lester. "Humanitarianism and Empire: New Research Agendas." *Journal of Imperial and Commonwealth History* 40, no. 5 (2012): 729–747.

Stokke, Olav ed. Western Middle Powers and Global Poverty: The Determinants of the Aid Policies of Canada, Denmark, the Netherlands. Uppsala: Scandinavian Institute of African Studies, 1989.

Thörn, Håkan. Anti-Apartheid and the Emergence of a Global Civil Society. Basingstoke: Palgrave Macmillan, 2006.

"Trócaire Launches Annual Lenten Campaign." One World, April 1982.

Watenpaugh, Keith. *Bread from Stones: The Middle East and the Making of Modern Humanitarianism*. Oakland, CA: University of California Press, 2015.

Woolcock, Michael, Simon Szreter, and Vijayendra Rao. "How and Why History Matters for Development Policy." In *History, Historians and Development Policy: A Necessary Dialogue*, edited by CA Bayly, Vijayendra Rao, Simon Szreter and Michael Woolcock, 3–38. Manchester: Manchester University Press, 2011.